Author's response to reviews

Title: Absence of mutations in NR2E1 and SNX3 in five patients with MMEP (microcephaly, microphthalmia, ectrodactyly, and prognathism) and related phenotypes

Authors:

Ravinesh A Kumar (rakumar@cmmt.ubc.ca)
David B Everman (deverman@ggc.org)
Chad T Morgan (morgan114@charter.net)
Anne Slavotinek (slavotia@peds.ucsf.edu)
Charles E Schwartz (ceschwartz@ggc.org)
Elizabeth M Simpson (simpson@cmmt.ubc.ca)

Version: 4 Date: 20 July 2007

Author's response to reviews: see over
July 19th, 2007

The BioMed Central Editorial Team
Tel: +44 (0)20 7631 9921
Facsimile: +44 (0)20 7631 9923
E-mail: editorial@biomedcentral.com

Re: Absence of mutations in NR2E1 and SNX3 in five patients with MMEP (microcephaly, microphthalmia, ectrodactyly, and prognathism) and related phenotypes

I am delighted to learn of your formal acceptance of our manuscript for publication as a Research Article in *BMC Medical Genetics*.

As per your request we have made the following formatting changes:

- Header has been removed
- ‘Authors’ contribution’ has been changed to ‘Authors’ contributions’
- Section headings have been converted to sentence case

Please note that the figures appear to have the correct orientation on the upload website but do not have the correct orientation in the links within the acceptance email sent July 14th, 2007. I have left a message with the editorial production team about this and will follow up with them to ensure the orientation is correct.

I trust the manuscript is now formatted suitably for publication.

Yours sincerely,

Elizabeth M. Simpson, Ph.D.